

RESEARCH OUTPUTS / RÉSULTATS DE RECHERCHE

The impact of lockdown on air pollution

Huang, Linyuan; Xie, Rui; Yang, Guohao

Published in: China Economic Review

DOI: 10.1016/j.chieco.2021.101731

Publication date: 2022

Document Version Publisher's PDF, also known as Version of record

Link to publication

Citation for pulished version (HARVARD): Huang, L, Xie, R & Yang, G 2022, 'The impact of lockdown on air pollution: Evidence from an instrument', *China Economic Review*, vol. 71, 101731. https://doi.org/10.1016/j.chieco.2021.101731

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
 You may not further distribute the material or use it for any profit-making activity or commercial gain
- You may freely distribute the URL identifying the publication in the public portal ?

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect





China Economic Review

journal homepage: www.elsevier.com/locate/chieco

The impact of lockdown on air pollution: Evidence from an instrument



Linyuan Huang^a, Rui Xie^{a,*}, Guohao Yang^b

^a Hunan University, Changsha, School of Economics and Trade, Hunan 410079, China

^b University of Namur, Centre of Research in the Economics of Development (CRED), Rue de Bruxelles 61, 5000 Namur, Belgium

ARTICLE INFO

Keywords: COVID-19 Lockdown Environment Health China

ABSTRACT

This paper studies the impact of lockdown measures in response to the outbreak of COVID-19 on a prefecture's air pollution in China. To avoid potential endogenous problems, we exploit the bilateral population flow from the Baidu Migration Index to predict prefectures' probability to undertake lockdown measures. Our results using difference-in-differences with the instrumental variable show that a prefecture's lockdown measures significantly reduce its air quality index (*AQI*) by around 35%, and yet the result for difference-in-differences with OLS is only around 11%. We also find that a prefecture under lockdown reduces its PM_{10} and $PM_{2.5}$ by around 25% and 35% respectively, and the results of diff-in-diff with OLS are only around 11% and 12%. The sharp difference between these two approaches seems to imply that there is a strong heterogeneity in lockdown stringency across prefectures.

1. Introduction

The outbreak of the highly infectious coronavirus COVID-19 has had a substantial impact on human health. In order to curb its spread, many local governments in China undertook lockdown measures which substantially limited the mobility of the population and the operation of factories and other businesses. This study contributes to our understanding of the impact of local Chinese prefectures' lockdown measures on their air quality, and explores the potential channel in air quality changes. To do so, we exploit the policy responses of the local governments in the face of the COVID-19 outbreak as a quasi-natural experiment. China provides a good case study because COVID-19 first broke out in China near the end of 2019, and more than one third of Chinese prefectures undertook massive, unprecedented lockdown measures to curb the spread of the disease. Moreover, industrial development in China is associated with a substantial increase in air pollution, which poses a threat to public health. The problem of air pollution in China receives considerable attention in academia (Chen, Jin, Kumar, & Shi, 2013; Greenstone, He, Jia, & Liu, 2020; He, Fan, & Zhou, 2016). To conduct our analysis, we employ a difference-in-differences (henceforth diff-in-diff) identification strategy to estimate the effect of a prefecture's lockdown measures on its air quality. Then, we use the rich information of the lockdown documents from local office to discuss the heterogeneity in the effects of lockdown measures. Finally, by collecting the most recent air quality data from monitoring Stations, we are able to check the long-term effect of lockdown measures.

Due to our high frequency (daily) data, we can compare the average change of a prefecture's air quality precisely before and after a

https://doi.org/10.1016/j.chieco.2021.101731

Received 13 October 2020; Received in revised form 21 September 2021; Accepted 4 December 2021 Available online 14 December 2021 1043-951X/© 2021 Elsevier Inc. All rights reserved.

^{*} Corresponding author at: Hunan University, School of Economics and Trade, 109 Shijia Chong Road, Yuelu District, Changsha City, Hunan Province, China.

E-mail addresses: huanglinyuan@hnu.edu.cn (L. Huang), xrxrui@126.com (R. Xie), guohao.yang@unamur.be (G. Yang).

L. Huang et al.

prefecture's lockdown. The key challenge to our identification strategy is that both a prefecture's decision to undertake and the timing of undertaking a lockdown might not be random. We consider the political concerns of the local government. There are two potential concerns that might affect a local political leader's decision to implement lockdown measures. The first one is economic. A lockdown measure implies shutting down restaurants, hotels, and factories, which slows the growth of the local economy considerably. Given that a politician's promotion opportunity is closely linked with economic growth (Jia, Kudamatsu, & Seim, 2015; Li & Zhou, 2005), he or she might be reluctant to enact such measures until the situation is quite bad. The second concern is that a local politician could be replaced if the spread of COVID-19 gets out of control.¹

To address this issue, we construct a prefecture's time-varying risk of COVID-19 outbreak, conditional on the average inflow of population from epidemic areas (other prefectures). This measure can be used to predict a prefecture's decision regarding lockdown measures. Our identification strategy relies on the fact that if a prefecture receives a significant inflow of population from an epidemic area, this prefecture might face a high risk of COVID-19 outbreak and should undertake lockdown measures (Fang, Wang, & Yang, 2020).² To construct our instrument, we use a prefecture's population inflow from another prefecture p', to capture the relative importance of population flow from p' to p between January 1st and January 9th.³ Before the outbreak of COVID-19, we believe the local governments had no incentives to manipulate this population flow out of the COVID-19 concerns. Therefore, this population flow could be treated as exogenous. To construct the time-varying aspect of a prefecture's COVID-19 risk, we use a time-varying dummy to indicate whether prefecture p' had more than a certain number of confirmed cases \overline{C} at time t. The combination of these two variables allows us to predict whether a prefecture received a high inflow of population from a COVID-19 epidemic area, and whether this prefecture should have undertaken a lockdown.

We first use a diff-in-diff identification strategy to estimate the impact of lockdown on the local air pollution with our instrumental variable. What we identified with this instrumental variable is the effect of a prefecture's lockdown measures on air pollution if the lockdown measures were caused by the risk. Our estimates show that a prefecture's lockdown approach improves its air condition significantly. From the estimate of diff-in-diff with 2 stage-least-square (2SLS), a prefecture under lockdown reduces its air quality index by around 35%, and the estimate of diff-in-diff with OLS is only around 11%. We also find that a prefecture under lockdown reduces its PM_{10} and $PM_{2.5}$ by around 25% and 35% respectively, and the results of diff-in-diff with OLS are only around 11% and 12%. The sharp difference between diff-in-diff with 2SLS and with OLS seems to suggest that there exists substantial heterogeneity in the stringency of lockdown implementation. This further suggests that the prefectures with a high risk of COVID-19 as predicted by our instrument did not undertake stringent lockdown measures, and prefectures with a low risk of COVID-19 did undertake stringent lockdown measures. This could also be confirmed by the fact that the accumulative confirmed cases of every prefecture located in Inner Mongolia and Liaoning province totaled less than 10 until the end of March. Both Inner Mongolia and Liaoning province declared a lockdown in every prefecture in their provinces on February 10.

Given that a prefecture's lockdown significantly reduces the density of air pollutants, we still need to carefully explain the mechanism behind. A prefecture's lockdown measures limit both the mobility of a prefecture's population and the operation of its factories. Only a handful of companies in industries such as retail selling of medical supplies, medical services, and utilities were allowed to operate during the lockdown. We explore two major sources of a prefecture's air pollution: traffic and manufacturing production. To assess the effects of a traffic reduction on air pollution due to lockdown, we construct a prefecture's road intensity to capture its potential traffic volume in 2017. To capture the effects of production on pollution, we construct a prefecture's aggregate manufacturing firms' coal consumption intensity. A prefecture's aggregate manufacturing coal consumption intensity represents the dirtiness of this prefecture's productions. In other words, a prefecture is expected to experience a stronger improvement in air quality during its lockdown if it uses dirtier technology. Our estimates suggest that such prefectures experienced a stronger improvement in air conditions under lockdown, and that road intensity did not contribute significantly to the effects of lockdown on air quality.

We also investigated the heterogeneity in the effects of the various lockdown measures. By collecting the official documents about lockdown measures, we obtained the rich information of the severity of lockdown in different prefectures. For example, some low risk regions such as Yinchuan and Wuzhong tended to adopt weaker regulations like setting up checkpoints, cleaning and disinfecting the community, canceling mass gatherings and so on. Therefore, we can conduct a series of regressions to check the heterogeneous effects of various lockdown measures. we find that there are significant differences among the effects of various lockdown measures on AQI.

We provide a brief discussion herein on how our study is linked with the literature. Our paper employs daily population migration to predict a prefecture's lockdown decision, and is in close relation to the public events and migration Fang et al. (2020); Qiu, Chen, and Shi (2020); Chinazzi et al. (2020); Gray and Mueller (2012); Lu, Bengtsson, and Holme (2012).⁴ Our study also relates to the literature about public environmental policy and its implications for the environment (Chen, Ebenstein, Greenstone, & Li, 2013; Davis, 2008; Viard & Fu, 2015). In addition, some recent researches studied the impacts of COVID-19 on air pollution (Dai, Hou, Liu, Zhang, & Feng, 2021; Dang & Trinh, 2021; He, Pan, & Tanaka, 2020), which are close to our research topic. We will discuss our differences and

¹ Coronavirus: Beijing purges Communist Party heads in Hubei over "botched" outbreak response in provincial capital of Wuhan https://www.scmp.com/news/china/politics/article/3050372/coronavirus-beijings-purge-over-virus-takes-down-top-communist.

² The inflow of population from epidemic areas has a significant impact on the outbreak of COVID-19 in a prefecture outside Hubei. In general, the first infected case in a prefecture outside Hubei had a traveling history to Wuhan before he/she was confirmed.

³ The population flow from prefecture p' to p between January 1st and January 9th captures the relative importance of the population flow of prefecture p' to p during the pandemic.

⁴ The public events in the literature referred to the epidemic, flooding, earthquake, etc.

L. Huang et al.

contributions in detail in the literature review.

The remainder of the paper is organized as follows. Section 2 introduces the background of COVID-19. Section 3 reviews the related literature. Section 4 and Section 5 describe our empirical strategy and the data sources, respectively. Section 6 presents the estimation results. Section 7 offers concluding remarks.

2. Background

2.1. The outbreak of COVID-19

The World Health Organization (WHO) has declared a global pandemic over a new coronavirus which causes an illness officially known as COVID-19. This disease has quickly spread around the world in the past months. Coronaviruses (CoV) are a large family of viruses that cause illness ranging from the common cold to more severe diseases such as Middle East Respiratory Syndrome (MERS-CoV) and Severe Acute Respiratory Syndrome (SARS-CoV). A novel coronavirus (COVID-19) is a new strain that was discovered in 2019 and has not been previously identified in humans. Therefore, human body did not have immunity against this disease and vaccine had not been developed yet. Based on the studies of 41 Chinese patients Liu, Gayle, Wilder-Smith, and Rocklov (2020) suggest that the basic reproduction number (R0) for COVID-19 in 12 studies has a mean of 3.28 and a median of 2.79. This means that the COVID-19 is highly infectious, and one infected individual may lead three additional people to be infected on average. The findings in Huang et al. (2020) suggest that 41 patients hospitalized show high rates of respiratory distress, intensive care admission, and abnormal findings on chest computer tomography (CT) from December 2, 2019 to January 2, 2020 in Wuhan, and the death rate is 15%. Therefore, it is necessary for the government to undertake progressive measures to reduce the contact between people during the outbreak of the disease.

The epicenter in China, Wuhan, was the first prefecture to undertake a lockdown measure on January 23, 2020. The first infected case of COVID-19 was discovered in Wuhan, an inland port city with more than 11 million populations. Due to the highly infectious nature, the total number of infected cases increased tremendously. On January 23 of 2020, the central government announced to lock down the whole Wuhan city, when the documented accumulative confirmed cases in Wuhan was close to 500. This action included shutting down the cross-prefecture highway networks, ferries, high-speed railways and bus-line from and to Wuhan. Within Wuhan, restaurants, hotels, and the entertaining facilities were closed down. In the residence community in Wuhan, visitors were not allowed to go in. The residents living inside the community were only allowed to go outside three times a week for purchasing necessary groceries. These actions are believed to reduce people's contact, hence the contagion of the COVID-19. After Wuhan announced lockdown on January 23, other large prefectures outside Hubei province such as Wenzhou, Ningbo, Zhumadian, Hangzhou, Zhengzhou, Haerbin, and Fuzhou undertook similar measures, as they saw a rapidly rising number of infected cases.⁵ We also observe prefectures undertaking lockdown with no more than five accumulative COVID-19 confirmed cases after the COVID-19 period. These prefectures include Yingkou, Benxi, Liaoyang, and Wuhai.

The COVID-19 has caused substantially detrimental impact on both the health of the Chinese people and local economy. Up to October 1st, 2020, the total deaths of COVID-19 in China exceed 4000 and the total number of infected cases exceed 90,000.⁶ In response to the outbreak of this disease, 129 prefectures including which of Hubei (the total number of prefectures in China is 326) received a lockdown measure. Fig. A1 shows us the spatial distribution of lockdowns across different prefectures. The epicenter Wuhan located inside Hubei province, and a large number of prefectures with lockdowns located geographically close to the Hubei province. Other prefectures with lockdowns lie along the Northeastern part of China, such as prefectures from Inner Mongolia and Liaoning. The lockdown measures undertaken by the prefectures outside Hubei province undertook lockdown measures. Fig. A2 shows us the timing of the lockdowns by the prefectures. The prefectures from Hubei province undertook lockdown measures. Most of the prefectures undertook lockdown measures. Most of the prefectures undertook lockdown measures around February 1. Around February 5, 82 prefectures announced to undertake lockdown measures. Due to the lockdown measures around 300 million people had been affected, which consist of one fifth of the total population in China. In the meantime, around 210 prefectures announced to shut down the cross-prefectures' bus-line, ferries, and its own public transport services within the prefecture.

2.2. Dynamics of population flow and confirmed cases

2.2.1. Lockdown and population flow

What determines the lockdown policy? In our opinion, whether to take lockdown measure mainly depends on the potential threat of COVID-19. Unlike the previous epidemic, the COVID-19 spread around the world quickly in a few days and has a high mortality. In addition, the COVID-19 broke out in the lunar new year holiday, which is the most active period of migrant population in China. After China's authorities confirmed that the virus can pass from person to person on January 20, the serious threat of epidemic increased the motivation of officers to lockdown the prefecture to contain the virus's spread. Although the prefecture's lockdown timing was affected by a lot of other factors including the development of local economy, the proportion of travelers choosing public transportation, its

⁵ Wenzhou was locked down on February 2, 2020; Ningbo, Zhumadian, Hangzhou, Zhengzhou, Haerbin, and Fuzhou were locked down on February 4, 2020.

⁶ http://www.nhc.gov.cn/cms-search/xxgk/getManuscriptXxgk.htm?id=08c7b372eece40b3acaa46f03c738f69.

real-time resources and the local official's taste and ability to deal with the public health event. From the timing of lockdown and the characteristics of the prefectures with lockdown, we can conclude that the prefecture's exposure to the COVID-19 risk is the major factor to promote the lockdown decisions. In addition, under the conditions of the limited data, this factor is the most appropriate for constructing an exogenous instrumental variable.

The lockdown measure substantially limited the mobility of the people within a prefecture. According to the Baidu migration data, a prefecture's mobility intensity which captures the flow of population within the prefecture experienced a stronger reduction for the prefectures undertaking lockdown measures during the Spring Festival which lasted from January 23 to February 9 in China. Fig. A3 shows us the dynamics of mobility intensity in prefecture groups with and without lockdown measures. We split the prefectures into two groups depending on their choice of undertaking lockdown measures before February 29. We also define a group of prefectures with lockdown measures as treatment group and all the other prefectures as control group. From our definition, the prefectures inside the treatment or control group does not vary over time. From this figure, the treatment group, which is in solid line, had a relatively high mobility intensity before January 23. After January 23, the mobility intensity dropped for prefectures in both the treatment and control groups, and the prefectures in the treatment group experienced a much stronger reduction, which means that the mobility of population inside the prefecture was more restrictive for the treatment than the control group.

The lockdown measures also limit flow of population across prefectures. According to the daily immigration and outmigration scale of prefectures from the Baidu data, the migration scale reduced substantially during the outbreak of the COVID-19 in China. Fig. A4 presents the dynamics of the population flow across different prefectures for 2019 and 2020. The two upper panels in this figure report the average outflow (left) and inflow (right) in 2020. The two upper panels show that both the daily outflow and inflow of population experienced a reduction in our study period. The reduction of both outflow and inflow of population is much sharper for the prefectures with lockdown measures than the ones without these measures. As a comparison, we show the inflow and outflow of population flow for both prefecture groups in the lower panels of Fig. A4. From these lower panels, the dynamics of population flow for both prefecture groups are similar.

2.2.2. Dynamics of the confirmed case

To estimate the effects of a lockdown on the COVID-19 related health outcomes, we employ our framework which is discussed fully in the previous section. The data of COVID-19 outcomes comes from the National Health Commission of China, which reports the accumulative confirmed cases, deaths and recovery. We are interested in the number of daily new cases, and hence we calculate a prefecture's daily new COVID-19 related outcomes by taking the difference of the accumulative outcomes between two days. Fig. A5 shows us the dynamics of the daily new confirmed cases across different prefectures in China outside Hubei province. This figure shows that the average daily new case peaks at the beginning of February, before the massive lockdown measures took place.

3. Related literature

This paper exploits the population migration between the epicenter and a prefecture to predict the optimal lockdown action of this prefecture. Our paper is in close relation to the public events and migration. Fang et al. (2020) and Qiu, Chen, & Shi (2020) use the outmigration from the epicenter in China, Wuhan, to a prefecture, to predict this prefecture's infected cases. Chinazzi et al. (2020) propose to use a Global Epidemic and Mobility model to study the impact of travel limitation on the national and international spread of the COVID-19. Gray and Mueller (2012) investigate the consequences of climate-related natural disasters for long-term population mobility in rural Bangladesh, and find that flooding had modest effects on mobility that are most visible at moderate intensities and for women and the poor. Lu et al. (2012) analyze the movements of 1.9 million mobile phone users during the period from 42 days before, to 341 days after the devastating Haiti earthquake of January 12, 2010. Their findings suggest that population movements during disasters may be significantly more predictable than previously thought. Our study is different from the literature in the sense that we use the population migration from the epicenter to predict the optimal lockdown action outside the epicenter.

Our study is also related to the literature about public policy and its implication for the environment. Viard and Fu (2015) evaluate the pollution and labor supply reductions from Beijing's driving restrictions. Beijing's driving restrictions reduced air pollution but at the cost of less work time by those with discretionary labor supply. Chen, Ebenstein, et al. (2013) show that an arbitrary Chinese policy that greatly increases total suspended particulates (TSPs) air pollution caused the 500 million residents of Northern China to lose more than 2.5 billion life years of life expectancy. Chen, Jin, et al. (2013) show that the actions from the Chinese government to reduce traffic flow and production during and a little after the Beijing Olympic Games significantly improved the air quality. Almond, Chen, Greenstone, and Li (2009) find that the heating policy led to dramatically higher pollution levels in the North China. Davis (2008) shows that the travel restrictions in Mexico have no effect on improving air quality. Our identification is different from the literature in the sense that we address the potential endogenous problem of a lockdown by the local government using the daily bilateral migration information in China. Our identification strategy substantially alleviates the potential political concern to the decision of a lockdown.

The most relevant works to our paper are the researches of He et al. (2020), Dang and Trinh (2021) and Dai et al. (2021). He et al. (2020) used the DID approach to examine the short-term impacts of COVID-19 on air pollution in China. Dang and Trinh (2021) offer an early assessment with cross-national evidence on the causal impacts of COVID-19 on air pollution by using a regression discontinuity design approach. And Dai et al. (2021) applied machine learning to quantify the impacts of the COVID-19 lockdown as well as the Chinese Spring Festival (CSF) holidays on the air quality changes in 31 major Chinese cities. Our paper joints the discussion of the environmental impact of COVID-19 by contributing in the followings: (1) Introducing a new instrumental variable to address the potential endogenous concerns. In order to address the endogenous concerns, we exploit the population flow from the epidemic area to construct an instrument variable, which could also capture the unobservable COVID-19 risk from the perspective of the local

governments. Our results suggest massive difference in estimates with and without our IV, which seems to confirm our guess. (2) We also examine the longer-term effects of COVID-19 lockdown, which is not shown in the previous paper.

4. Empirical strategy

In this section, we propose to use a diff-in-diff identification strategy to identify the impacts of a prefecture's lockdown. The challenge of identification is that a prefecture's lockdown decision may not be random, because local politicians are concerned with their promotion opportunities, which are closely linked with local economic performance (Li & Zhou, 2005). If a politician undertakes radical isolation measures, he or she is aware that such measures may substantially slow the growth of the economy. Therefore, local politicians may have no incentive to impose lockdown measures. Additionally, we find that prefectures under lockdown were not the ones with the highest number of cases. For example, each prefecture in Liaoning province, with no more than 10 confirmed cases, undertook lockdown measures. Yet the average confirmed cases across different prefectures outside Hubei was around 40 as of the end of March, far higher than 10. Therefore, we expect to observe substantial heterogeneity with respect to the stringency of lockdown measures across prefectures.

To address this issue, we predict a prefecture's lockdown decision outside Hubei according to inflow of population from prefectures with an epidemic. Fang et al. (2020) suggests that a prefecture's inflow of population from Wuhan contributes significantly to its accumulative number of confirmed COVID-19 cases. We extended this measure by first identifying a prefecture p' to be an epidemic area at time t if its accumulative number of confirmed cases is higher than a certain threshold \overline{C} . A prefecture is subject to high risk of COVID-19 outbreak and should undertake lockdown measures if it receives a substantial inflow of population from epidemic areas. To capture the bilateral population flow from one prefecture p' to another one p and avoid the potential endogenous problem, we use their population flow two weeks before Wuhan's lockdown which marked the outbreak of COVID-19 in China (January 23rd). This bilateral population flow is meant to capture the daily commuting flow across different prefectures. We do not use a place's real-time population inflow from an epidemic area because the real-time population flow is likely to be affected by other COVID-19 related measures such as highway closures or the shutting down of cross-prefecture bus lines. Therefore, our instrumental variable (IV) is as follows:

$$\widehat{M}_{pt} = \sum_{t' < t} \sum_{p'} I(Case_{p't'} \ge \overline{C}) M_{p'p,t_0}$$

where $I(Case_{p't'} \ge \overline{C})$ is a dummy variable equal to 1 if the number of confirmed cases of prefecture p at time t is larger than \overline{C} , $M_{p'p, t0}$ captures the average bilateral flow of population from prefecture p to prefecture p between January 1 and January 9, 2020, and \widehat{M}_{pt} captures the potential threat of COVID-19 via population inflow. Another problem here is that \widehat{M}_{pt} becomes larger as date index t

increases. So we think of using the daily average $\frac{\widehat{M}_{Pl}}{t}$ as the final instrument.

The potential COVID-19 threat to prefecture *p* depends on the past inflow of population from prefecture *p*' and its confirmed cases. $M_{p'p, t0}$ captures the relative importance for *p* of population flow from *p*'. If there was a large population flow from *p*' to *p* before the outbreak of COVID-19, prefecture *p*' is expected to have a non-negligible population flow to *p* after the outbreak of COVID-19. Here, we assume that a prefecture's lockdown decision does not affect its past population inflow, which is captured by $M_{p'p, t0}$. Before January 9, a prefecture outside Hubei was less incentivized to manipulate population inflow. This is because January 9th is one day before the beginning of "Chunyun," the Spring Festival travel season, when there is a large population flow from one prefecture to another for the celebration and reunion.

Politicians outside of Hubei may have been less aware of the detrimental effects of COVID-19 on people's health and the economy before the Spring Festival. Also, prefecture p's lockdown decision might not have affected the accumulative number of confirmed cases in another prefecture p'. Our instrumental variable might also capture the relative stringency of lockdown measures, if more stringent measures were implemented in prefectures with higher risk of a COVID-19 outbreak.

Using a prefecture's risk of a COVID-19 outbreak, we estimated the following equation to generate a prediction for a prefecture's lockdown decision, which is also the first stage of our 2SLS estimation:

$$I(event_{pt} = 1) = \beta_0 + \frac{\beta_1 \widehat{M}_{pt}}{t} + \alpha_t + \alpha_p + \epsilon_{pt}$$
(1)

where $I(event_{pt} = 1)$ is a dummy variable that takes a value of 1 if prefecture p undertook lockdown measures at date t. With this approach, we could predict the probability that prefecture p underwent a lockdown depending on the potential threat of COVID-19 via population inflow.

Our instrumental variable allowed us to capture both the location and timing of a lockdown measure, depending on population inflow. If there was a past non-negligible flow of population from the potential epidemic area to prefecture p, prefecture p is expected to undertake lockdown measures. If the number of confirmed cases increases rapidly in the potential epidemic area, such that it quickly exceeds the threshold number C, this prefecture p is expected to implement earlier lockdown measures. We use the Baidu Migration Index to measure the population flow from one prefecture to another. The Baidu Migration Index provides daily information on the bilateral traveling population across different prefectures in China.

With the predicted probability of a lockdown, we adopted a diff-in-diff approach to evaluate the impact of lockdown measures on the environment. We used a dummy variable to measure whether a prefecture announced lockdown measures at time *t*. If prefecture *p*

undertook lockdown measures on date t, then it kept these measures till the end, implying March 31 of 2020 in our study.

$$ln(y_{pt}) = \alpha_p + \alpha_t + \gamma_0 \widehat{I}(event_{pt} = 1) + \epsilon_{pt}$$
⁽²⁾

where y_{pt} represents the outcome of interest for prefecture p at date t, α_p represents the prefecture fixed effects, and α_t represents the time fixed effects.

According to Imbens (2014), a valid instrumental variable has to satisfy two criteria: (1) The instrument is correlated with the endogenous variable, and (2) the instrument affects the dependent variable only indirectly, also known as the exclusion restriction. The correlation between the endogenous variable and the instrumental variable has been discussed in our construction process and will be tested statistically by conducting a first-stage F-test, which will been shown in Section 6. As for the exclusion restriction, it is hard to test whether or not the instrument is uncorrelated with the error term. We argue that potential threat of COVID-19 via population inflow between January 1 and January 9, 2020 is not directly related to the air quality of prefectures, but has the indirect effect by influencing the prefecture's lockdown decision. Given that the confirmed cases outside a prefecture can hardly be correlated with the air of this prefecture, the main concern for our exclusion restriction is that the transportation supplied for the population flow will affect a prefecture's environment. To deal with the problem, we adopted the past average inflow of population between January 1 and January 9, 2020 to capture the potential threat of COVID-19, which would not affect the transport activities later. Another evidence to support the exclusion restriction is that we find that the within-prefecture migration in prefectures with lockdown and which without lockdown had no significant difference. So our instrument do not affect our outcomes through channels other than the lockdown. Nevertheless, there are still some limitations in our construction. First, the accumulated flows between other prefecture with COVID-19 cases are not equivalent to COVID-19 risk. Second, we cannot control other factors which could affect the lockdown decision. Indeed, our instrument is not fully equal to the COVID-19 risk, and it is hard to measure other determinants due to the lack of daily data, but all we need for identification is a reliable exogenous variation. And our instrument could serve as a valid instrument to capture the potential risk of COVID-19, although it is not perfect.

5. Data

5.1. Documents from local offices

The first dataset we use comes from the official documents that report measures of the local government in response to the outbreak of the COVID-19. We manually collect all the related documents from the local offices. These documents report the date when they announce to undertake the lockdown measures. In addition, these documents also report specific measures, such as shutting down the public transport services and curfews. We have collected documents for around 130 different prefectures, which officially announced lockdown measures. The list of Locked-down prefectures is reported in Table A1. Table A1 also describes the stringency of lockdown measures undertaken by prefectures, which can help to discuss the heterogeneity in the effect of lockdown measures on air pollution.

5.2. Annual environmental survey of polluting firms (AESPF)

The data on firms' pollution emissions comes from the Annual Environmental Survey of Polluting Firms (AESPF) of China. It includes rich information on firms' environmental performance such as main pollutants emissions (waste water, industry exhaust, *COD*, *SO*₂, soot and dust, etc.), pollution abatement (sulfur dioxide removal, soot removal, desulfurization device, etc.), and energy consumption (fuel coal, fuel oil, raw coal and industrial fresh water, etc.), among others. Since the "Tenth Five- Year Plan" in 2001, the scope, frequency, main indicators and reporting methods of the environmental survey have basically become stabilized. As long as a firm's discharged pollutants are in the top 85% of the total pollutant discharge at the county level, this firm is included in the key-point environmental survey list, and it must send a unified environmental report to the environmental authorities every year. After review and verification by relevant administrative departments, these data are entered into the database. This article uses the main fuel indicators in the database.

5.3. Population migration data

We obtain daily bilateral prefecture population flows from Baidu Migration, a electronic navigating map run by the largest Chinese search engine, Baidu. Baidu can map the migration trajectory of mobile phone users by analyzing the location information of mobile phone users through the Location Based Service (LBS) open platform. There are currently more than 500 million mobile Internet users in China, and Baidu's LBS open platform's location services cover hundreds of thousands of apps, and the number of location requests per day exceeds billions. So the location information data provided by Baidu Migration is undoubtedly the most convincing in China. The Baidu Migration data set covers 120,142 pairs of prefectures per day for 364 Chinese prefectures between January 12 and March 31 in 2019, and between January 1 and March 31 in 2020. Note that, by the lunar calendar, the data covers the same period of 24 days before and 54 days after the Chinese New Year, respectively for year 2019 and year 2020. In addition, Baidu offers information on the daily population mobility intensity within the prefecture, and this is a panel sample with more than 20 thousand observations each year.

Summary statistics of data.

	All prefectures	2020 (01. January - 3	31. March)		2019 (12. January -	13. April)
		Treatment group		Control group	Treatment group	After Treatment
		Before Treatment	After Treatment			
Panel A: Environment data						
401	73.04	89.34	61.07	73.36	81.10	82.11
AQI	(48.46)	(55.96)	(25.49)	(51.44)	(45.74)	(56.12)
PM	45.98	64.05	35.52	45.33	54.50	52.26
r 1v12.5	(40.47)	(47.94)	(21.54)	(42.15)	(39.58)	(48.22)
PM-	70.12	81.66	59.15	71.23	84.39	90.06
1 14110	(72.71)	(56.49)	(35.29)	(84.52)	(51.01)	(103.02)
Panel B: Health data						
Confirmed cases	27.75	9.23	74.28	15.75	-	-
	(77.29)	(40.48)	(142.06)	(31.67)	-	-
New confirmed cores	0.49	1.33	0.54	0.26	-	-
New commed cases	(2.98)	(5.75)	(2.56)	(1.91)	-	-
Cured cases	18.73	0.24	53.51	10.89	-	-
Curcu cases	(59.21)	(1.98)	(110.79)	(26.04)	-	-
New cured cases	0.48	0.06	1.34	0.27	-	-
New Curcu cases	(2.29)	(0.48)	(4.15)	(1.36)	-	-
Panel C: Migration data						
Within prefecture migration	3.92	4.47	3.81	3.83	4.71	4.43
within-prefecture ingration	(1.34)	(1.55)	(1.22)	(1.29)	(0.76)	(0.78)
Outflow	0.75	1.64	0.71	0.55	1.80	0.89
Ouniow	(1.32)	(2.83)	(0.78)	(0.64)	(2.33)	(0.88)
Inflow	0.75	1.45	0.80	0.56	1.83	0.87
milow	(1.02)	(1.63)	(1.09)	(0.66)	(2.39)	(0.88)
Prefectures	314.00	117.00	117.00	197.00	117.00	197.00

Notes: This table mainly reports the mean values and standard deviations of different variables at the daily level.

5.4. Pollution and weather data

Daily air quality data were collected from the records of 1650 local monitoring stations and each monitor reports the air quality index and intensity of pollutants in the air. This dataset reports the concentration of coarse particulate matter (PM_{10}), fine particulate matter ($PM_{2.5}$), carbon monoxide (CO), sulfur dioxide (SO_2) and nitrogen oxides (NO_X). A general concern is that the local government in China has incentives to manipulate the local air quality data because the performance of the local air quality affects substantially his or her promotion possibility (Ghanem & Zhang, 2014). However, this concern has been significantly alleviated as the China upgraded its air quality monitoring system to measure pollutants more precisely. This newly adopted monitoring system gathers pollutant sample automatically and at the same time reports the results. So the newly adopted system significantly improved the air quality as shown by Greenstone et al. (2020), because this monitoring system makes it very difficult for the local government to manipulate the air pollution data. Our air quality data covers all the monitoring stations in China. if a prefecture has multiple monitoring stations, we will take the arithmetical mean of them as the prefecture's outcome.

Table 1 reports the summary statistics for the air pollution, cases and migration data. As seen from the Panel A, the lockdown prefectures were, on average, more polluted than the control prefectures before the lockdowns. And we can see a sharp decline in all air pollutants concentrations due to the lockdown. From the Panel B, we found that the new confirmed cases experienced a decline and the new cured cases gone up, on average, after the lockdown. Panel C shows that the lockdown prefectures experienced a strong reduction in both the within-prefecture migration and cross-prefecture migration.

6. Empirical results

6.1. Estimation results for the environment

6.1.1. Estimates with diff-in-diff

The threat of COVID-19 on a prefecture depends on how we define an epidemic area, which links with the value of \overline{C} . The value of \overline{C} determines prefecture *p*'s risk of COVID-19 exposure. If we choose a relatively lower value for \overline{C} , a larger number of prefectures would be considered epidemic areas. This implies that a prefecture with population inflow from many different prefectures has a high risk of exposure to COVID-19. On the other hand, if we choose a relatively higher value for \overline{C} , a smaller number of prefectures will be considered epidemic areas. This implies that a prefecture with population inflow from many different prefectures will be considered epidemic areas. This implies that a prefecture with population inflow from many different prefectures could balance out the risk of COVID-19. As shown in the appendix, our estimates are robust to a reasonable variation of \overline{C} .

We find that the average local accumulative number of confirmed cases on the day when the lockdown started is around 48. Therefore, we tried several values, namely 5, 10, 30, 50, 80, and 100, for \overline{C} to run the first stage. We show our results for the second

L. Huang et a

The impact of a prefecture's lockdown on the environment.

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Results in the firs	t stage					
Dependent variable: I(event _{pt} =	= 1)					
\widehat{M}_{pt5}	0.00193***					
Frie	(0, 00)					
Â	(0.00)	0 00202***				
101 pt,10		0.00202				
â		(0.00)				
$M_{pt,30}$			0.00230***			
			(0.00)			
$\widehat{M}_{pt,50}$				0.00276***		
				(0.00)		
$\widehat{M}_{ m pt \ 80}$					0.00341***	
pi,oo					(0,00)	
\widehat{M} , 100					(0.00)	0 00408***
<i>wipt</i> ,100						0.00400
Moothor controls	Vac	Vac	Vee	Vac	Vee	(0.00)
Brefecture fixed effects	Vec	Vec	Vec	Vec	Vec	Vec
Date-fixed effects	Ves	Yes	Ves	Ves	Ves	Ves
B-squared	0.687	0.685	0.681	0.680	0.678	0.677
F-value	51.48	52.21	42.46	35.50	29.76	20.52
Observations	28,574	28,574	28,574	28,574	28,574	28,574
			,	,	,	,
Panel B: Results for AQI, P	M_{10} and $PM_{2.5}$ in the	second stage	(0)		(5)	(6)
Denondont worighter	(1)	(2)	(3)	(4)	(\mathbf{D})	(0)
<i>K</i> event = 1)	III(AQI)	III(AQI)	$M(PM_{10})$	III(PM10)	III(PM2.5)	0.24E**
I(evenupt = 1)	-0.109	-0.346	-0.109	-0.245	-0.110	-0.343
FSt	OLS	251.5	OLS	251.5	OLS	2515
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $I(event_{pt} = 1) = 1$ if a prefecture p undertook a lockdown at day t. Otherwise, $I(event_{pt} = 1) = 0$. $\widehat{M}_{pt,\overline{C}}$ refers to the threat of COVID-19 of prefecture p at day t with $\overline{C} \in \{5, 10, 30, 50, 80, 100\}$. We choose 30 for the value of \overline{C} in the second stage. ln (AQI) represents the log value of a prefecture's air quality index, $ln(PM_{10})$ represents the log value of a prefecture's fine particulate matters ($\mu g/m^3$).

stage estimation with \overline{C} , and the results are robust if we choose \overline{C} from 5 to 100. Panel A of Table 2 reports the results for all the threshold values of \overline{C} , and they are all significant and positive. Column (1) of this table shows that a one-unit increase in prefecture p's threat of COVID-19 is associated with a 0.2% increase in this prefecture's probability of announcing a lockdown. As we increase the threshold value for \overline{C} , the significance does not change, and the correlation between a prefecture's threat and its lockdown behavior increases. Column (6) shows that a one-unit increase in a prefecture's threat of COVID-19 is 0.4% associated with a prefecture's lockdown decision, when we choose 100 for the value of \overline{C} .

Table 2 reports the estimates of the first stage in panel A and the estimates of the second stage for PM_{10} , $PM_{2.5}$, and AQI in panel B. Here, we choose 30 for the value of \overline{C} , which is close to the average accumulative confirmed cases as of the end of March. Column (2) shows that a prefecture's lockdown measures reduced its AQI by around 35% with diff-in-diff IV, and column (1) shows that a prefecture's lockdown measures reduced its AQI by around 11% with diff-in-diff with OLS. Column (4) suggests that a prefecture's lockdown reduced its density of PM_{10} by around 25%. A prefecture's Lockdown reduced its density of $PM_{2.5}$ by around 35%, which is similar to the effects on AQI.

So, what can we learn from this exercise? The estimates of diff-in-diff with OLS treat every prefecture as the same as long as this prefecture announced a lockdown with an official document. In the OLS regression, what we estimated is the simple correlation between a prefecture's lockdown and air condition, with which we could not interpret as causal. But in the 2SLS regression, we adopted an instrument to capture the relative stringency of lockdown among the prefectures according to the COVID-19 risk, which can reflect the intensity of policy implementation. When a prefecture faced with higher risk, the local government had a stronger incentive to strengthen the enforcement of lockdown. Our results have confirmed this argument. The estimates of 2SLS were much larger than the estimates of OLS, suggesting that a prefecture with a higher risk of COVID-19 is associated with a more substantial reduction in air pollution after lockdown. In *Section 6.3.2*, we provide estimates with the same framework but taking into account the log value of real-time population inflow. The estimates are robust.

The potential channels that contribute to the effects of a lockdown on the environment.

	(1)	(2)	(3)	(4)
Dependent variables:	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)
$I(event_{pt} = 1)$	-0.348**	-0.887*	0.458*	0.159
-	(0.16)	(0.51)	(0.24)	(0.46)
$I(event_{pt} = 1) \times Road_{pt_0}^{den}$		0.36		0.16
A A U		(0.23)		(0.20)
$I(event_{pt} = 1) \times Coal_{pt_0'}^{den}$			-2.21***	-2.45***
· · ·			(0.49)	(0.36)
$I(event_{pt} = 1) \times Population_{pt_0'}$		0.00	-0.00	-0.00
		(0.00)	(0.00)	(0.00)
Est.	IV	IV	IV	IV
Weather controls	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $I(event_{pt} = 1) = 1$ if a prefecture p undertook a lockdown at day t. Otherwise, $I(event_{pt} = 1) = 0$. $Road_{pt0'}^{den}$ refers to the initial road intensity of prefecture p, $Coal_{pt0'}^{den}$ refers to the average initial manufacture firms' coal consumption intensity of prefecture p.

6.1.2. Potential mechanism

Given that a prefecture's lockdown substantially reduces its air pollution, we examine the possible role of manufacturing production and traffic. Chen, Ebenstein, et al. (2013) show that limiting the flow of daily traffic and the shut-down of manufacturing plants contributed substantially to Beijing's air quality during the 2008 Olympic Games. Following this idea, we use a prefecture's average coal consumption intensity, i.e., the ratio between coal consumption and sales, in 2010 from AESPF to measure the dirtiness of production at the prefecture level.⁷ If a prefecture has a relatively higher coal consumption intensity, this means that the production of this prefecture relies heavily on coal as fuel in production. To capture the effect of reduced traffic due to lockdown measures, we use a prefecture's road intensity in 2017, i.e., the ratio between a prefecture's total road length and its population. To implement this idea, we estimate the following equation:

$$ln(y_{pt}) = \alpha_p + \alpha_t + \xi_0 \widehat{I}(event_{pt} = 1) + \xi_1 \widehat{I}(event_{pt} = 1) \times Road_{pt'}^{den} + \xi_2 \widehat{I}(event_{pt} = 1) \times Coal_{pt'}^{den} + \epsilon_{pt}$$
(3)

where $Road_{pt_0'}^{den}$ captures the road density of prefecture p at the initial period t_0' and $Coal_{pt_0'}^{den}$ captures the coal intensity of the manufacturing firms of prefecture p at time t_0' . We expect both the estimates for ξ_1 and ξ_2 to be negative and significant. A prefecture with dirtier production technology or a high volume of traffic is expected to experience a greater reduction in pollution once under lockdown.

Table 3 reports the results of the estimates, controlling for the size of a prefecture in terms of its population. These estimates help us understand the extent to which traffic and manufacturing production contribute to the effects of a lockdown on the environment. Column (2) shows that a prefecture with a high initial level of road intensity does not have a stronger reduction in pollution. Column (3) shows that manufacturing firms' coal consumption intensity within a prefecture absorbs nearly all the effects of that prefecture's lockdown on the environment, indicating that the effects of a prefecture's lockdown are due to a reduction in manufacturing production. This argument can be confirmed if we put all the interaction terms into one regression as shown in column (4). In column (4), only the interaction term between coal consumption intensity and the lockdown dummy is significantly negative.

6.2. The heterogeneity in the lockdown measure

It's important to note that the stringency of lockdown was totally different from the heterogeneity in the lockdown measure. The former could not be observed directly but could be taken into account in our 2SLS regressions. The latter could be found in the official documents about lockdown. We demonstrated the specific lockdown measures in different prefectures in Table A1. Beyond those common measures, including setting up checkpoints, cleaning and disinfecting the community, canceling mass gatherings, the 117 prefectures adopted different actions to prevent the virus from further spreading. Because these measures are not comparable and heterogenous, we cannot treat them as the variable treatment intensities. Thus the identification strategy with variable treatment intensity (Angrist & Imbens, 1995) is not suitable for our question.

Even so, the rich information of official documents can help us investigate the heterogeneity of different measure's effect on air pollution. Hence we conducted a set of regressions by changing the dependent variable in first stage for checking the heterogeneous effects of various lockdown measures. The results are represented in Table 4, where $I(measure_{ipt} = 1)$ is a dummy variable that takes a value of 1 if prefecture p undertook the lockdown *measure_i* at date t. We found that there are significant differences among the effects of

⁷ We assume that the relative dirtiness of technology in 2010 could reflect the relative dirtiness of technology today.

Table 4	
The heterogeneous effects of different lockdown measures.	

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variables:	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)
$I(measure_{ipt} = 1)$	-0.366*** (0.108)	-0.522*** (0.188)	-0.498*** (0.167)	-0.505*** (0.179)	-1.074^{**} (0.521)	-0.354*** (0.109)	-0.531*** (0.159)	-0.572*** (0.194)	-0.374*** (0.114)	-0.746*** (0.256)	-1.602 (1.145)	-1.118** (0.482)
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects Observations	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. *I(measure*_{ipt} = 1) = 1 if a prefecture p undertook a *measure*_i at day t. Otherwise *I(measure*_{ipt} = 1) = 0. The specific lockdown measures are as follows: 1) Contactless delivery; 2) Communities with confirmed cases may implement closed isolation; 3) All visitors and exotic vehicles are not allowed to enter the community; 4) Travelers from the Epidemic Area Need to be Quarantined for 14 Days upon Arrival to the Prefecture; 5) Close contacts need to be quarantined at assembly sites; 6) Public places not necessary for the lives of residents should be closed; 7) Suspension of non-essential project; 8) Communities should strengthen management of rental housing; 9) Persons Stranded in the Epidemic Area Could not Return; 10) Health QR code or pass shall be used to manage people; 11) Wedding suspended and funeral simplified; 12) Each family (exclude home quarantine families) could only assign one family member to purchase daily supplies every two/three days.

_ . . .

(6) In(CO) 0.00201 (0.0242) Yes Yes Yes 111,360

Table 5

The long-term effect of the le	he long-term effect of the lockdown measure.								
	(1)	(2)	(3)	(4)	(5)				
Dependent variables:	ln(AQI)	ln(<i>PM</i> ₁₀)	ln(PM _{2.5})	$\ln(SO_2)$	$ln(NO_2)$				
$I(event_p = 1)$	0.0268	0.0286	0.0479	0.0791*	0.103***				
*	(0.0305)	(0.0378)	(0.0376)	(0.0437)	(0.0373)				
Weather controls	Yes	Yes	Yes	Yes	Yes				
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes				
Date-fixed effects	Yes	Yes	Yes	Yes	Yes				
Observations	111,360	111,360	111,360	111,360	111,360				

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation.

Table 6

The impact of a lockdown on the environment (levels).

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	AQI	AQI	PM10	PM10	PM _{2.5}	PM _{2.5}
$I(event_{pt} = 1)$	-10.95***	-42.21**	-14.77***	-45.34**	-7.07**	-27.44*
	(3.60)	(18.87)	(4.53)	(20.80)	(2.77)	(14.60)
Est.	OLS	2SLS	OLS	2SLS	OLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. *AQI* measures the absolute value of a prefecture's air quality index, PM_{10} measures the absolute value of a prefecture's coarse particulate matters density ($\mu g/m^3$), $PM_{2.5}$ measures the absolute value of a prefecture's fine particulate matters density ($\mu g/m^3$). Weather controls include daily temperature, its square, humidity, and precipitation.

various lockdown measures on AQI. Among these measures, $measure_{12}$ had the biggest positive effect on air pollution, which reduced the AQI by around 112%. The results were correspondent with our expectation. Because the $measure_{12}$ stipulated that each family (exclude home quarantine families) could only assign one family member to purchase daily supplies every two/three days, and other citizens were not allowed to go out except for medical treatment, epidemic prevention and control, and working with the approval of local government. This regulation severely restricted the movement of people and production, and thus could reduce air pollution more significantly compared with other measures.

6.3. The long-term effect of the lockdown measure

When the COVID-19 epidemic started to slow down in China in late February, most prefectures started to cancel the lockdown measures. Although lockdown is a short-run Decision, whether it will cause a long-lasting effect excites our interest. In order to analyze this problem, we extended our sample period to March 31, 2021. Because all the prefectures cancelled the lockdown policy after March of 2020, what we can identify is just the coefficient of a dummy, which equal to 1 if a prefecture ever took the lockdown measure. Table 5 reports the results on the long-term effect of the lockdown measure. In the long run, the effect of lockdown on air quality has faded away. This suggests that the lockdown only improve the air quality in the short-term.

6.4. More robustness checks

6.4.1. Event study

We first evaluate the impacts of a prefecture's lockdown measure using an event study framework. A government is identified to undertake a lockdown approach if it declared to adopt a lockdown measure via an official document. To assess the impact, we compare the trends of the prefectures with lockdown measures to those without lockdown measures in the following specification:

$$ln(y_{pt}) = \alpha_p + \alpha_t + \sum_{j=-30}^{30} \beta_j \widehat{I}(event_{pt} = 1) + \epsilon_{pt}$$

This specification includes prefecture (α_p) and time (α_t) fixed effects. Standard errors are clustered at the prefecture level where the shocks generally happen.

Fig. A6 displays the results of the event study about environment graphically. The first panel in Fig. A6 shows us the results of log of *AQI*. Before the event, our results suggest that the prefectures with lockdown are systematically more polluted. After a prefecture undertook a lockdown, the estimates for the effects of lockdown on *AQI* became more apparent. After the first week, the estimates are

China Economic Review 71 (2022) 101731

Table 7

Robustness check by including the samples of 2019.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	ln(AQI)	ln(AQI)	$\overline{ln(PM_{10})}$	$ln(PM_{10})$	ln(PM _{2.5})	$ln(PM_{2.5})$
$I(event_{pt} = 1)$	-0.348**	-0.392***	-0.245*	-0.388***	-0.345**	-0.475***
*	(0.16)	(0.13)	(0.13)	(0.14)	(0.18)	(0.15)
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	57,148	28,574	57,148	28,574	57,148
Year Covered	2020	2019, 2020	2020	2019, 2020	2020	2019, 2020

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. "Year Covered" refers to the sample year we use to estimate our results.

Table 8

Robustness check by including the actual inflow scale as a control variable.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	ln(AQI)	ln(AQI)	$ln(PM_{10})$	$ln(PM_{10})$	ln(PM _{2.5})	ln(PM _{2.5})
$I(event_{pt} = 1)$	-0.348**	-0.465***	-0.245*	-0.523***	-0.345**	-0.362**
	(0.16)	(0.14)	(0.13)	(0.16)	(0.18)	(0.15)
$ln(M_{p,t})$		0.099***		0.149***		0.129***
•		(0.02)		(0.02)		(0.02)
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $ln(M_{p_i}, t)$ represents the log value of total population inflow of prefecture p at day t.

significant and negative, which implies that a prefecture's lockdown improve the air condition significantly. The results of PM_{10} and $PM_{2.5}$, shown in the second and the third panel, are similar to the results of AQI.

6.4.2. Robustness checks

6.4.2.1. The effect of a prefecture's lockdown on the absolute value of the pollutants. Table 6 reports the estimates of a prefecture's lockdown on the absolute value of air pollution. The results show that a prefecture's lockdown reduces its AQI, PM_{10} and $PM_{2.5}$ by around 42 units, 45 units and 27 units using diff-in-diff with 2SLS, respectively. The results using diff-in-diff with OLS are less than one-third of which with 2SLS for all of these variables. These are consistent with what we find in the relative changes.

6.4.2.2. We also include the samples of 2019 into the regression. Given that our estimates using the time period from January 1 to March 31 of 2020, which marks the end of COVID-19 in China. We use different time periods to compare our estimate to check whether our approaches are sensitive to timing. In order to do this, we first try January 1 of 2020 to March 31 of 2020. The reason why we use this time period is that the daily new confirmed cases of prefectures outside Hubei almost fell to 0 at the end of March. In another setting, we also include the same period in 2019. In this setting, we expand our control group. Since COVID-19 did not happen in the same period of 2019, we manually set a prefecture risk of COVID-19 to be 0. Table 7 reports these estimates. The columns in odd number include only the samples from 2020, and the columns in even number include samples both from 2020 and 2019. The estimates for the three dependent variables do not vary too much between two samples, as shown in the table.

6.4.2.3. We also include the log value of actual inflow scale as a control variable. The sharp difference between the estimates using 2SLS and OLS indicates the heterogeneous effect of lockdown on the environment. Nevertheless, there is still concern that the exclusion assumption of our IV is not satisfied because that the inflow of population from epidemic areas might affect pollution other than lockdown. Since it is not possible to test this exclusion assumption directly, we control for the log value of actual inflow of population in our regression and check whether our estimate is robust. Table 8 reports these results. The columns in even number of this table report the results adding this control variable. Our estimate shows that adding this control variable does not change our diff-in-diff with 2SLS estimates too much.

6.4.2.4. We estimate the effects of a prefecture's lockdown with various \overline{C} . Given that we only use $\overline{C} = 30$ in our main text, it is tempting to know how our estimates would be affected if different \overline{C} is chosen. To check whether our results are robust to different values of \overline{C} ,

Robustness check by choosing different \overline{C} .

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)	ln(AQI)
$I(event_{pt} = 1)$	-0.386**	-0.382^{**}	-0.348**	-0.291*	-0.331*	-0.464**
	(0.15)	(0.15)	(0.16)	(0.17)	(0.19)	(0.23)
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574	28,574	28,574
C	5	10	30	50	80	100

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. \overline{C} refers to the critical value in identifying a prefecture as an epidemic area.

Table 10

Results of the new \overline{C} which depends on the newly-added confirmed cases.

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Results in the first sta	ge					
Dependent variables: $I(event_{pt} = 1)$						
$\widehat{M}_{pt,3}$	0.00483***					
x '	(0.00)					
Â		0 00699***				
<i>p</i> i,6		(0,00)				
ŵ		(0.00)	0.000(0+++			
<i>W</i> _{pt,10}			0.00960***			
<u>^</u>			(0.00)			
M _{pt,15}				0.0141***		
				(0.00)		
$\widehat{M}_{pt,25}$					0.0244***	
					(0.00)	
$\widehat{M}_{\rm pf}$ 30						0.0399***
<i>pi</i> ,50						(0.00)
Weather controls	Ves	Ves	Ves	Ves	Ves	(0.00) Yes
Prefecture-fixed effects	Yes	Yes	Ves	Ves	Ves	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
B-squared	0.680	0.677	0.674	0.675	0.674	0.675
F-value	38.67	27 41	17.82	21.47	27.04	36.68
Observations	28.574	28.574	28,574	28.574	28.574	28,574
Danal B: Paculte for AOL DM	and PM. in the second	nd stage	- ,		- ,	-,
Tanci D. Results for Mor, TM10	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables	ln(AOI)	ln(AOI)	$ln(PM_{10})$	$ln(PM_{10})$	$ln(PM_{2})$	$ln(PM_{2})$
$I(event_{ret} = 1)$	-0.109***	-0.429***	-0.109***	-0.429***	-0.116***	-0.329**
-contrapp -y	(0.00)	(0.16)	(0.00)	(0.14)	(0.00)	(0.15)
ESt.	OLS	2SLS	OLS	2SLS	OLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $\widehat{M}_{pt,\overline{C}}$ refers to the threat of COVID-19 of prefecture p at day t with $\overline{C} \in \{3, 6, 10, 15, 25, 30\}$, which is measured by the newly-added confirmed cases.

we use all the possible values suggested in the main text, from 5 to 100, to implement our diff-in-diff with 2SLS framework. Table 9 reports the results. Our results suggest that our estimates do not vary too much for different values of \overline{C} .

6.4.2.5. \overline{C} depends on the newly-added confirmed cases instead of accumulative cases. In the main results, we identify a prefecture p' to be an epidemic area at time t if its accumulative number of confirmed cases is higher than a certain threshold \overline{C} . In order to check the robustness of the empirical findings, we turn to use the newly-added confirmed cases to measure the risk of a prefecture. Since the average newly-added confirmed cases on the day the lockdown starts is around 3. We chose several values, namely 3, 6, 10, 15, 25, and 30, for new \overline{C} to run the first stage. And we also show the results for the second stage estimation with $\overline{C} = 3$ in Table 10. The results are robust not only in the first stage but also in the second stage.

The impact of a prefecture's lockdown on the environment (include Hubei).

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Results in the first	stage					
Dependent variables: I(event _{pt} =	= 1)					
$\widehat{M}_{pt,5}$	0.00190***					
	(0.00)					
Â		0.00200***				
101 pt,10		(0.00200				
<u>.</u>		(0.00)				
$M_{pt,30}$			0.00230***			
			(0.00)			
$\widehat{M}_{pt,50}$				0.00283***		
				(0.00)		
$\widehat{M}_{nt,80}$					0.00347***	
					(0.00)	
\widehat{M} , 100					(0000)	0 00409***
191 pt,100						0.00405
Mooth on controls	Vac	Vee	Vac	Vac	Vac	(0.00)
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Data fixed affects	Yes	Yes	Vec	Vec	Vec	Ves
Date-fixed effects	0.601	0.60	165	165	0.695	0.694
E voluo	0.091 E1 02	0.09 E2.42	0.007 4E 1E	40.72	24.92	0.064
Observations	20 484	20 484	45.15	40.72	20 494	24.05
Observations	29,404	29,404	29,404	29,404	29,404	29,404
Panel B: Results for AQI, PI	M ₁₀ and PM _{2.5} in the	second stage				
	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	ln(AQI)	ln(AQI)	$ln(PM_{10})$	$ln(PM_{10})$	$ln(PM_{2.5})$	$ln(PM_{2.5})$
$I(event_{pt} = 1)$	-0.109***	-0.441***	-0.109***	-0.447***	-0.116^{***}	-0.349**
	(0.00)	(0.15)	(0.00)	(0.14)	(0.00)	(0.15)
ESt.	OLS	2SLS	OLS	2SLS	OLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	29,484	29,484	29,484	29,484	29,484	29,484

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $\hat{M}_{pt,\overline{C}}$ refers to the threat of COVID-19 of prefecture p at day t with $\overline{C} \in \{5, 10, 30, 50, 80, 100\}$. We choose 30 for the value of \overline{C} in the second stage.

Table 12

Robustness check for other air pollutant.

	(1)	(2)	(3)	(4)
Dependent variables:	ln(SO ₂)	ln(NO ₂)	ln(CO)	$ln(O_3)$
$I(event_{pt} = 1)$	-0.468***	-0.339**	-0.191*	0.202*
-	(0.14)	(0.13)	(0.10)	(0.12)
Est.	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. All the dependent variables are the log values.

6.4.2.6. We also include the samples of Hubei province. In the main results, we eliminated the samples of Hubei province because it is the epicenter of COVID-19, and the preventive and control measures of Hubei might be fundamentally different from which of other regions. Therefore we're concerned about that the samples of Hubei may form the interference to our estimates. But the lockdown of Hubei province seems to be a more exogenous shock to identify the impact on air pollution. So we provide the related regression results which include the samples of Hubei province in Table 11. These results are close to the main results when we excluded the samples of Hubei province.

6.4.2.7. We also test the results for other air-quality outcome measures. In the baseline regressions, we focus on the impacts of the lockdown on the main indicators of air quality. In order to test whether the results are robust to other air pollutant, we adopt the same method to investigate the effect of lockdown on sulfur dioxide (SO_2), nitrogen dioxide (NO_2), carbon monoxide (CO) and ozone (O_3).

The results of spatial dependence.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variables:	ln(AQI)	$ln(PM_{10})$	$ln(PM_{2.5})$	ln(AQI)	$ln(PM_{10})$	ln(PM _{2.5})
$I(event_{pt} = 1)$	-0.390**	-0.301***	-0.377**	-0.96**	-0.80**	-0.97**
$I(event_{p't} = 1)$	(0.15) -0.32***	(0.12) -0.36***	(0.16) -0.21***	(0.37)	(0.33)	(0.39)
	(0.06)	(0.06)	(0.06)			
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	28,574	28,574	28,574	8526	8526	8526

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $I(event_{p't} = 1) = 1$ if a prefecture p's nearby prefecture undertook a lockdown at day t. Otherwise, $I(event_{p't} = 1) = 0$.

Table 14

Robustness check about the prefecture's exposure to the COVID-19 risk.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Dependent variables:	ln(AQI)	$ln(PM_{10})$	$ln(PM_{2.5})$	ln(AQI)	$ln(PM_{10})$	$ln(PM_{2.5})$	ln(AQI)	$ln(PM_{10})$	$ln(PM_{2.5})$
$I(event_{pt} = 1)$	-0.522^{**} (0.20)	-0.412^{**} (0.17)	-0.501^{**} (0.21)	-0.297** (0.12)	-0.238** (0.09)	-0.268^{**} (0.13)	-0.103*** (0.00)	-0.101*** (0.00)	-0.117^{***} (0.00)
$ln(M_{p, t})$							-0.000** (0.00)	-0.000*** (0.00)	-0.000 (0.00)
Est.	2SLS	2SLS	2SLS	2SLS	2SLS	2SLS	OLS	OLS	OLS
Weather controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Prefecture-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Date-fixed effects Observations	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574	Yes 28,574

Notes: Standard errors are in parentheses and clustered at the prefecture level. *** p < 0.01, ** p < 0.05, * p < 0.1. Weather controls include daily temperature, its square, humidity, and precipitation. $ln(M_{p_i}, t)$ represents the log value of total population inflow of prefecture p at day t.

The results are shown in Table 12. We find that lockdown measure reduces all the pollutants except for O3. This may be caused by the significant reduction of $PM_{2.5}$ and NO_2 , which could inhibit the chemical formation of ozone.

6.4.2.8. Test the spatial dependence. Although we have discussed the exclusion restriction for the instrument variable, there is still spatial dimension that cannot be ignored. Suppose the pollutants can reach the nearby prefecture in a few hours, it may become a possible channel through which \hat{M}_{pt} is correlated with air quality in *p*. So we hope to test the possible correlation. First, we control for a dummy which equal to 1 when a nearby prefecture adopts lockdown measures, the results of which are shown in columns (1–3) of Table 13. Second, we eliminate the samples whose neighbor adopts the lockdown measures to run the related regression, which can be seen from columns (4–6) of Table 13. Compare the results of Table 13 with the baseline results, we find that we might underestimated the effect of lockdown on air pollution without considering the spatial dependence.

6.4.2.9. Some questions about the prefecture's exposure to the COVID-19 risk. In our baseline model, for excluding the impact of the Spring Festival travel rush on the instrument, we only used the average bilateral flow of population between January 1 and January 9, 2020 to capture the potential threat of COVID-19. As days go by, this prefecture's exposure to the risk from prefecture p' keeps rising no matter prefecture p' has locked down or not. To solve this problem, we try to use the average population flow of longer time to predict the prefecture's exposure to the COVID-19 risk in consideration of decreasing flow after the breakout of COVID-19. The columns (1–3) of Table 14 shows that when we use the average bilateral flow of population between January 1 and February 12, 2020 to predict the potential threat of COVID-19, the estimates do not vary too much compared with the baseline results. However, there is still some concern that if a prefecture's connected area p' has confirmed cases over \overline{C} , whether prefecture p' has an earlier lockdown or not will affect the prefecture differently (Fang et al. (2020); Qiu, Chen, & Shi (2020)). Thus we adjust our construction of \hat{M}_{pt} by turning $I(Case_{p't'} \ge \overline{C})$ into $I(Case_{p't'} \ge \overline{C})I(event_{p't} = 0)$, which is a dummy variable equal to 1 if the number of confirmed cases of prefecture p' at date t is larger than \overline{C} and the prefecture p' did not undertake lockdown measures at date t. As shown in the columns (4–6) of Table 14, the estimates are smaller than which of baseline results but they are still bigger than estimates of OLS results. Moreover, we also try the OLS regressions controlling for real population inflow from an epidemic area in lag days shown in the columns (7–9) of Table 14. Adding this control variable does not change the estimates too much. The absolute values of estimates are still smaller than which of the 2SLS estimates.

7. Conclusion

This paper studies the effect of government response to COVID-19 on the environment. The high frequency (daily) data allows us to precisely compare changes in environmental quality across different prefectures before and after any prefecture undertook lockdown measures. In particular, we find that (1) a prefecture's lockdown had a substantial and significant clean-up effect on air quality, (2) the effect of the lockdown on the environment may be due to the reduction in manufacturing production, and (3) there are significant differences among the effects of various lockdown measures on air pollution.

To estimate the effects of a lockdown on the environment, we exploit the high frequency of the data to develop a novel identification strategy which helps to predict the behavior of the government. As high frequency data become more available, our approach provides an example of how to make the best use of daily variations to identify the effects of the public events and the government decisions. Although our instrument is limited because it is hard to completely prove the validity of it, it is the most appropriate instrument we can find in the available data.

Declaration of Competing Interest

None.

Acknowledgments

This research was generously supported by the Science Fund for Distinguished Young Scholars of Hunan province (Grant No. 2021JJ10027). We would like to thank the participants at CRED workshops. We owe special thanks to Albrecht Glitz for his guidance. We also want to express our thanks to Lorenzo Trimarchi, Eric Melander and Zhijie Xiao for their helpful comments and discussions.

Appendix

Table A1

The timing and lockdown measures of the prefectures

Beginning of Lockdown	Prefecture	m_1	m_2	<i>m</i> 3	<i>m</i> 4	m_5	<i>m</i> 6	<i>m</i> 7	<i>m</i> 8	<i>m</i> 9	<i>m</i> 10	<i>m</i> 11	<i>m</i> ₁₂
01 1 0000	Wuzhong	0	0	1	0	0	0	0	0	0	0	0	0
31 January 2020	Yinchuan	0	0	1	0	0	0	0	0	0	0	0	0
2 February 2020	Wenzhou	1	1	1	1	0	1	0	1	0	1	0	1
3 February 2020	Huaian	0	0	0	1	0	0	0	1	0	0	0	0
	Changzhou	1	1	1	1	0	0	0	1	0	0	1	0
	Fuzhou	1	1	0	1	0	1	0	1	1	0	1	1
	Haerbin	1	1	1	0	0	0	0	1	0	0	1	1
	Hangzhou	1	1	0	0	0	1	0	1	1	0	0	0
	Jingdezhen	1	1	1	1	0	1	0	1	1	1	0	1
	Linyi	1	1	0	1	0	1	0	1	0	0	1	1
4 February 2020	Nanjing	1	1	0	1	0	1	1	1	1	0	0	0
	Nantong	1	1	1	1	1	1	0	1	1	0	1	1
	Ningbo	1	0	1	0	0	1	0	1	1	0	1	1
	Xuzhou	1	1	1	0	0	0	1	1	0	0	1	1
	Zhengzhou	1	0	1	0	0	0	0	0	0	1	0	0
	Zhenjiang	1	0	1	1	0	0	1	0	1	1	0	0
	Zhumadian	0	0	1	1	1	0	0	1	1	0	0	0
	Haikou	1	1	0	1	0	1	0	1	0	1	1	0
	Hefei	1	1	0	1	1	1	1	1	0	0	1	1
	Jinan	1	1	0	0	0	0	0	1	0	0	1	0
	Kunming	1	1	0	1	0	1	1	1	0	0	0	0
	Nanchang	1	1	1	1	0	1	0	1	0	1	1	1
	Nanning	0	1	0	1	0	0	0	1	1	0	0	0
	Qingdao	1	1	0	1	1	0	0	1	0	1	1	0
5 February 2020	Rizhao	0	0	0	1	1	0	0	1	0	0	0	0
	Sanya	1	1	0	0	1	1	1	1	1	0	1	1
	Shijiazhuang	1	1	0	0	1	1	0	0	1	0	0	0
	Suqian	1	1	1	1	0	1	0	1	0	0	1	0
	Taian	1	1	0	1	1	1	0	1	0	1	1	1
	Taizhou	1	0	0	1	1	0	1	0	0	0	0	1
	Yancheng	0	0	1	1	0	0	0	0	1	0	0	0
	Yangzhou	1	0	1	1	0	1	1	1	1	0	1	0
	Anshan	1	1	1	1	1	0	0	1	0	1	1	0
6 February 2020	Benxi	1	1	1	1	1	0	0	1	0	1	1	0
	Chaoyang	1	1	1	1	1	0	0	1	0	1	1	0
											(a a m ti		***

(continued on next page)

Table A1 (continued)

Beginning of Lockdown	Prefecture	m_1	m_2	m_3	m_4	m_5	m ₆	<i>m</i> ₇	m_8	<i>m</i> 9	<i>m</i> ₁₀	<i>m</i> ₁₁	<i>m</i> ₁₂
	Dalian	1	1	1	1	1	0	0	1	0	1	1	0
	Dandong	1	1	1	1	1	0	0	1	0	1	1	0
	Fushun	1	1	1	1	1	0	0	1	0	1	1	0
	Fuxin	1	1	1	1	1	0	0	1	0	1	1	0
	Fuzhou	1	1	1	1	0	1	0	1	0	1	0	1
	Ganzhou	1	1	1	1	0	1	0	1	0	1	0	1
	Huludao	1	1	1	1	1	0	0	1	0	1	1	0
	Jian	1	1	1	1	0	1	0	1	0	1	0	1
	Jiin	1	1	1	1	1	0	0	1	0	1	1	0
	Jinzhou	1	1	1	1	0	1	0	1	0	1	0	1
	Liaoyang	1	1	1	1	1	0	0	1	0	1	1	0
	Maanshan	1	1	0	1	0	1	1	1	Ő	0	0	0
	Maanshan	1	1	0	1	1	1	1	1	0	0	1	1
	Neijiang	0	1	0	1	1	1	0	1	0	0	1	0
	Panjin	1	1	1	1	1	0	0	1	0	1	1	0
	Pingxiang	1	1	1	1	0	1	0	1	0	1	0	1
	Shangrao	1	1	1	1	0	1	0	1	0	1	0	1
	Shenyang	1	1	1	1	1	0	0	1	0	1	1	0
	Suzhou	1	0	1	0	0	1	0	0	1	0	0	0
	Tieling	1	1	1	1	1	0	0	1	0	1	1	0
	Xinyu	1	1	1	1	0	1	0	1	0	1	0	1
	Yaan Vichup	1	0	1	1	0	1	0	1	0	0	1	1
	Vingkou	1	1	1	1	1	1	0	1	0	1	1	1
	Vingtan	1	1	1	1	0	1	0	1	0	1	0	1
	Zhuhai	1	0	0	1	Ő	0	0	1	1	0	0	0
	Anging	1	1	0	1	1	1	1	1	0	0	1	1
	Bengbu	1	1	0	1	1	1	1	1	0	0	1	1
	Bozhou	1	1	0	1	1	1	1	1	0	0	1	1
	Chengdu	1	1	0	1	0	0	1	1	0	0	0	0
	Chizhou	1	1	0	1	1	1	1	1	0	0	1	1
	Chuzhou	1	1	0	1	1	1	1	1	0	0	1	1
	Fuyang	1	1	0	1	1	1	1	1	0	0	1	1
	Guangyuan	1	1	0	1	1	0	1	1	0	0	1	1
	Guangzhou	1	0	0	1	0	1	0	1	1	1	0	0
	Guiyang	1	1	0	1	1	0	1	1	0	0	0	0
	Huaipen	1	1	0	1	1	1	1	1	0	0	1	1
7 February 2020	Huangshan	1	1	0	1	1	1	1	1	0	0	1	1
7 rebruary 2020	Lanzhou	1	0	0	0	0	1	1	1	0	0	0	0
	Lianvungang	1	0 0	0	1	1	1	0	1	1	0	1	1
	Luan	1	1	0	1	1	1	1	1	0	0	1	1
	Shenzhen	1	1	1	1	1	1	1	1	1	1	0	0
	Suining	1	1	0	1	0	0	1	1	1	0	1	1
	Suzhou	1	1	0	1	1	1	1	1	0	0	1	1
	Tangshan	1	1	0	1	0	1	0	1	0	0	1	0
	Tianjin	1	0	0	0	0	1	1	1	0	0	0	0
	Tongling	1	1	0	1	1	1	1	1	0	0	1	1
	Wuhu	1	1	0	1	1	1	1	1	0	0	1	1
	Zunui	1	1	0	1	1	1	1	1	0	0	1	1
	Chongging	1	1	1	1	1	1	1	1	0	0	0	1
8 February 2020	Eoshan	1	1	0	1	0	1	0	1	0	0	1	0
o rebruary 2020	Zivang	1	1	0	0	0	1	0	1	0	0	1	0
	Baicheng	0	1	0	1	Ő	1	0	1	Ő	0	0	0
	Baishan	0	1	0	1	0	1	0	1	0	0	0	0
	Changchun	0	1	0	1	0	1	0	1	0	0	0	0
	Deyang	1	1	1	0	0	1	1	1	1	0	0	0
	Dongguan	1	1	1	1	0	1	0	0	1	0	0	0
	Hanzhong	0	0	1	1	0	0	0	0	0	0	0	0
9 February 2020	Huizhou	1	1	0	1	0	1	0	1	1	0	0	0
	Liaoyuan	0	1	0	1	0	1	0	1	0	0	0	0
	Mianyang	1	1	0	1	0	1	1	1	0	0	0	0
	Siping	0	1	0	1	0	1	0	1	0	0	0	0
	Songyuan	0	1	0	1	0	1	0	1	0	0	0	0
	Tonghua	0	1	0	1	0	1	0	1	0	0	0	0
	WUXI Beijing	1	0	1	1	1	1	U 1	0	0	0	1	0
10 February 2020	Shanobai	1	0	0	1	1	0	0	0	1	0	0	0
	onungitai	-	5	0	1	+	5	5	0	*			0

(continued on next page)

Table A1 (continued)

Beginning of Lockdown	Prefecture	m_1	m_2	m_3	m_4	m_5	m_6	<i>m</i> 7	<i>m</i> 8	<i>m</i> 9	<i>m</i> ₁₀	<i>m</i> ₁₁	m_{12}
	Alashan	1	1	0	1	1	1	0	1	0	0	1	1
	Baotou	1	1	0	1	1	1	0	1	0	0	1	1
	Bayannaoer	1	1	0	1	1	1	0	1	0	0	1	1
	Chifeng	1	1	0	1	1	1	0	1	0	0	1	1
	Eerduosi	1	1	0	1	1	1	0	1	0	0	1	1
10 Eshmann 2020	Huhehaote	1	1	0	1	1	1	0	1	0	0	1	1
12 February 2020	Hulunbeier	1	1	0	1	1	1	0	1	0	0	1	1
	Tongliao	1	1	0	1	1	1	0	1	0	0	1	1
	Wuhai	1	1	0	1	1	1	0	1	0	0	1	1
	Wulanchabu	1	1	0	1	1	1	0	1	0	0	1	1
	Xilinguole	1	1	0	1	1	1	0	1	0	0	1	1
	Xingan	1	1	0	1	1	1	0	1	0	0	1	1

Notes: The number 1 represents that the prefecture adopts the corresponding lockdown measure. This table only shows the heterogenetic lockdown measures across prefectures. So the common measures such as setting up checkpoints, cleaning and disinfecting the community, canceling mass gatherings and so on, are not included in the table. Besides, the 12 prefectures of Hubei are not included in this table because their lockdown measures are much more severe than which of other prefectures. The specific lockdown measures are as follows: 1) Contactless delivery; 2) Communities with confirmed cases may implement closed isolation; 3)All visitors and exotic vehicles are not allowed to enter the community; 4) Travelers from the Epidemic Area Need to be Quarantined for 14 Days upon Arrival to the Prefecture; 5) Close contacts need to be quarantined at assembly sites; 6) Public places not necessary for the lives of residents should be closed; 7) Suspension of non-essential project; 8) Communities should strengthen management of rental housing; 9) Persons Stranded in the Epidemic Area Could not Return; 10) Health QR code or pass shall be used to manage people; 11) Wedding suspended and funeral simplified; 12) Each family (exclude home quarantine families) could only assign one family member to purchase daily supplies every two/three days.





Fig. A1. The Spatial distribution of lockdowns.

Notes: This figure shows us the spatial distribution of lockdown undertook by the prefecture. In the map, the yellow triangles represent prefectures of Hubei Province, the red triangles represent the prefectures with lockdowns, and the different depths of color of the blue reflect the different levels of the number of confirmed COVID-19 Cases as of Feb 29, 2020.



Fig. A2. The timing of the lockdowns.

Notes: This figure shows us the timing of the lockdowns. Between January 23 and January 28, 13 prefectures inside the Hubei province undertook lockdown measures. After January 30, prefectures outside Hubei started to *undertook* lockdowns around February 5th. The red line, which indicates the date January 25, 2020, marked the starting of the Spring Festival Holiday, and the other red line which indicates the date February 9th, marked the end of the Spring Festival.



Fig. A3. Average within-prefecture flow.

Notes: This figure shows us the dynamics of the population mobility within the prefecture. We divide the prefectures into two groups: with and without a lockdown measure. The solid line indicates the group of prefectures with lockdown measures and the dash line indicates the group of prefectures without lockdown measures. January 23rd marked the day when Wuhan undertook a lockdown measure, two days before Spring Festival. February 9th marked the end of the Spring Festival.



Fig. A4. The dynamics of population flow.

Notes: This figure shows us the dynamics of daily total population outflow and inflow for two different prefecture groups in 2019 and 2020 respectively. The solid line represents the prefectures that received lockdown measures (treatment group). The dash line represents the prefectures without lockdown measures (control group). The prefectures in both the group with and without lockdown measures are time-invariant. If a prefecture received a lockdown measure during our study period, this prefecture would be assigned in the treatment group. The shade area represents the Spring Festival period.





Notes: This figure plots the dynamics of the average accumulated number of infected cases, the average daily new infected cases and the four prefectures with most infected cases outside Hubei. The black solid and dash lines represent the average daily new confirmed cases and the average number of cumulative confirmed cases in China outside Hubei province, respectively. The dash purple, orange, light blue and green lines represent the number of cumulative cases in Beijing, Shenzhen, Wenzhou, and Chongqing respectively.



Fig. A6. Event study of lockdown on the air pollution.

Notes: This figure shows the results of the event study for air pollution. The y axis represents the log of AQI, PM10 and PM2.5, and the x axis represents time. w1, w2, and w3 represents the first, second and the third week after the lockdown, respectively. $w \ge 4$ represents four weeks after lockdown. w-1, w-2, and w-3 represents the first, second and the third week before the lockdown, respectively. $w \le -4$ represents four weeks before lockdown. Our estimates have already controlled for a prefecture's daily precipitation and temperature, which are both time-variant. We also report 90% confidence bound for the daily estimates. We omit the estimates of a day before the lockdown.

References

- Almond, D., Chen, Y., Greenstone, M., & Li, H. (2009). Winter heating or clean air? Unintended impacts of China's Huai river policy. American Economic Review, 99(2), 184–190.
- Angrist, J. D., & Imbens, G. W. (1995). Two-stage least squares estimation of average causal effects in models with variable treatment intensity. *Publications of the American Statistical Association, 90*(430), 431–442.

Chen, Y., Ebenstein, A., Greenstone, M., & Li, H. (2013). Evidence on the impact of sustained exposure to air pollution on life expectancy from China's Huai river policy. *Proceedings of the National Academy of Sciences*, 110(32), 12936–12941.

Chen, Y., Jin, G. Z., Kumar, N., & Shi, G. (2013). The promise of Beijing: Evaluating the impact of the 2008 Olympic Games on air quality. Journal of Environmental Economics and Management, 66(3), 424–443.

Chinazzi, M., Davis, J. T., Ajelli, M., Gioannini, C., Litvinova, M., Merler, S., ... Vespignani, A. (2020). The effect of travel restrictions on the spread of the 2019 novel coronavirus (covid-19) outbreak. *Science*, 368(6489), 395–400.

Dai, Q., Hou, L., Liu, B., Zhang, Y., & Feng, Y. (2021). Spring festival and covid lockdown: Disentangling pm sources in major Chinese cities. *Geophysical Research Letters*, 48(11).

Dang, H.-A. H., & Trinh, T.-A. (2021). Does the covid-19 lockdown improve global air quality? New cross-national evidence on its unintended consequences. Journal of Environmental Economics and Management, 105, Article 102401.

Davis, L. (2008). The effect of driving restrictions on air quality in Mexico city. Journal of Political Economy, 116, 38-81.

Fang, H., Wang, L., & Yang, Y. (2020). Human mobility restrictions and the spread of the novel coronavirus (2019-ncov) in China. Journal of Public Economics, 191, Article 104272.

Ghanem, D., & Zhang, J. (2014). Effortless perfection:' do Chinese cities manipulate air pollution data? Journal of Environmental Economics and Management, 68(2). Gray, C. L., & Mueller, V. (2012). Natural disasters and population mobility in Bangladesh. Proceedings of the National Academy of Sciences, 109(16).

Greenstone, M., He, G., Jia, R., & Liu, T. (2020). Can technology solve the principal-agent problem? Evidence from China's war on air pollution (No. w27502). National Bureau of Economic Research.

He, G., Fan, M., & Zhou, M. (2016). The effect of air pollution on mortality in China: Evidence from the 2008 Beijing Olympic games. Journal of Environmental Economics and Management, 79.

He, G., Pan, Y., & Tanaka, T. (2020). The short-term impacts of covid-19 lockdown on urban air pollution in China. Nature Sustainability, 3.

Huang, C., Wang, Y., Li, X., Ren, L., Zhao, J., Hu, Y., Zhang, L., Fan, G., Xu, J., Gu, X., Cheng, Z., Yu, T., Xia, J., Wei, Y., Wu, W., Xie, X., Yin, W., Li, H., Liu, M., Xiao, Y., Gao, H., Guo, L., Xie, J., Wang, G., Jiang, R., Gao, Z., Jin, Q., Wang, J., & Cao, B. (2020). Clinical features of patients infected with 2019 novel

coronavirus in Wuhan, China. Lancet, 395(10223), 497–506. https://doi.org/10.1016/S0140-6736(20)30183-5 Imbens, G. W. (2014). Instrumental variables: An Econometrician's perspective. Statistical Science, 29(3), 323–358.

Jia, R., Kudamatsu, M., & Seim, D. (2015). Political selection in China: The complementary roles of connections and performance. Journal of the European Economic Association, 13(4), 631–668.

Li, H., & Zhou, L.-A. (2005). Political turnover and economic performance: The incentive role of personnel control in China. Journal of Public Economics, 89(9), 1743–1762.

Liu, Y., Gayle, A. A., Wilder-Smith, A., & Rocklov, J. (2020). The reproductive number of COVID- 19 is higher compared to SARS coronavirus. Journal of Travel Medicine, 27(2).

Lu, X., Bengtsson, L., & Holme, P. (2012). Predictability of population displacement after the 2010 Haiti earthquake. Proceedings of the National Academy of Sciences, 109(29), 11576–11581.

Qiu, Y., Chen, X., & Shi, W. (2020). Impacts of social and economic factors on the transmission of coronavirus disease 2019 (COVID-19) in China. Journal of Population Economics, 33, 1127–1172. https://doi.org/10.1007/s00148-020-00778-2

Viard, V. B., & Fu, S. (2015). The effect of Beijing's driving restrictions on pollution and economic activity. Journal of Public Economics, 125, 98-115.